Author's response to reviews

Title: Mycophenolate mofetil modulates adhesion receptors of the beta1 integrin family on tumor cells: impact on tumor recurrence and malignancy

Authors:

Tobias Engl (engl@em.uni-frankfurt.de)
Jasmina Makarevic (jmakarevic@air-net.de)
Borna Relja (jmakarevic@air-net.de)
Iyad Natsheh (blaheta@em.uni-frankfurt.de)
Iris Muller (Iris.Mueller@em.uni-frankfurt.de)
Wolf-Dietrich Beecken (beecken@em.uni-frankfurt.de)
Dietger Jonas (beecken@em.uni-frankfurt.de)
Roman A Blaheta (blaheta@em.uni-frankfurt.de)

Version: 3 Date: 28 December 2004

All formatting changes requested have been done:

Capitalisation has been removed from the title and from all headings.

Colons have been removed from the abstract subheadings.

"Acknowledgements" have been moved to after the authors contributions.

"Competing interests" have been included.

Table 1 was divided into cells/fields.

Several figures have been cropped. Figures 2a and 2b are now figures 2 and 3. Figure legends have been changed accordingly.